Comparison of Treatment-emergent Resistance-associated Mutations Among Single-tablet Regimens and Cabotegravir + Rilpivirine for the Treatment of Virologically Suppressed People With HIV: A Systematic Literature Review and Network Meta-analysis

Ishfaq Rashid¹, Teerapon Dhippayom², Connor Willis¹, Howard Weston Schmutz¹, Moti Ramgopal³, Elizabeth Sherman⁴, Rachel Safran⁵, Nicholas Yared⁶, Amy R Weinberg⁷, Soodi Navadeh⁷, Nathan R Unger⁷, Nathorn Chaiyakunapruk¹

¹Department of Pharmacotherapy, University of Utah College of Pharmacy, Salt Lake City, UT, USA; ²Faculty of Pharmaceutical Sciences, Naresuan University, Phitsanulok, Thailand; ³Midway Specialty Care Center, Fort Pierce, FL, USA; ⁴Department of Pharmacy Practice, Nova Southeastern University College of Pharmacy, Davie, FL, USA; ⁶Department of Internal Medicine, MultiCare Health System INW, Spokane, WA, USA; ⁶Department of Medicine, Division of Infectious Diseases, Henry Ford Health System, Detroit, MI, USA; ⁷Gilead Sciences, Inc., Foster City, CA, USA

P051



Conclusions

- In virologically suppressed people with HIV, bictegravir/emtricitabine/tenofovir alafenamide (B/F/TAF) had the highest probability of preventing treatment-emergent resistance-associated mutations (TE-RAMs) and tended to have the lowest risk of TE-RAMs
- There was a trend in which cabotegravir + rilpivirine (CAB + RPV) dosed every 8 weeks (Q8W) had a higher risk of resistance compared to all integrase strand transfer inhibitor (INSTI)-based single-tablet regimens (STRs) and was not as protective as other 2-drug regimens, performing similarly to STRs with lower barriers to resistance
- The risk of discontinuation due to an adverse event was significantly higher for CAB + RPV than for the second-generation INSTI-based 3-drug STRs B/F/TAF and dolutegravir/abacavir/lamivudine
- Compared to staying on B/F/TAF, switching to CAB + RPV was associated with a significantly higher risk of discontinuation due to adverse events in addition to a likely increased risk of TE-RAMs, particularly for CAB + RPV O8W
- · Until there is a cure, well-tolerated and durable antiretroviral regimens are needed for long-term success; moreover, clinicians should include the differential risk of TE-RAMs in shared decision-making discussions when switching antiretroviral therapy in stable, suppressed individuals since resistance impacts current and future treatment options

Plain Language Summary

- Oral single-tablet regimens (medicines that have more than 1 drug in them) are effective for people with HIV who take their medicine every day. These medicines help keep the virus from being found in the blood and help to stop resistance (which can cause the medicine to no longer work)
- Cabotegravir + rilpivirine is another treatment choice, which is given by injection and does not have to be taken every day. However, some people may still develop resistance to the medicine, even when they get their injections regularly
- If a medicine no longer works due to resistance, it becomes harder to find other medicines to treat HIV
- This study compared differences between oral single-tablet regimens and injectable cabotegravir + rilpivirine, including the chance of people becoming resistant to or having to stop taking the medicine due to side effects
- Bictegravir/emtricitabine/tenofovir alafenamide (a single-tablet regimen) was predicted to be the best at preventing resistance in people with HIV. Cabotegravir + rilpivirine had similar results to those of other single-tablet regimens that tend to lead to resistance development and stop working
- · More people with HIV stopped taking cabotegravir + rilpivirine because of side effects compared to those taking bictegravir/emtricitabine/tenofovir alafenamide or dolutegravir/abacavir/lamivudine
- Switching to cabotegravir + rilpivirine was linked to a higher chance of stopping the medicine because of side effects. People who switched to cabotegravir + rilpivirine also had a higher chance of developing resistance

Introduction

- · For people with fully susceptible HIV, current guideline-preferred single-tablet regimens (STRs) with a high barrier to resistance achieve and maintain durable virologic suppression without the emergence of resistance in those who are adherent to their oral antiretroviral therapy (ART)¹⁻³
- Injectable cabotegravir + rilpivirine (CAB + RPV) provides an alternative method of administration for people with HIV (PWH), removing the need for daily oral therapy¹
- · Treatment-emergent resistance-associated mutations (TE-RAMs), including dual-class resistance, have been described in PWH who are adherent to the injection schedule for CAB + RPV: in 1 study, among 14 PWH who received CAB + RPV and had confirmed virologic failure, 7 developed dual-class TE-RAMs through 152 weeks despite receiving injections on time⁴
 - TE-RAMs complicate HIV control and management, limiting both current and future therapeutic options⁵
- This study compared the risk of TE-RAMs and discontinuation due to adverse events (AEs) among STRs and CAB + RPV in virologically suppressed PWH

Methods

- A systematic literature review was conducted for Phase 2, 3, and 4 randomised controlled trials (RCTs) that investigated switching to any STR or CAB + RPV with ≥48 weeks of follow-up in both arms and with results published from January 2003 to March 2024; studies were retrieved from PubMed, Embase, Cochrane CENTRAL, and EBSCO Open Dissertations
- Participants in the trials were virologically suppressed PWH aged ≥12 years
- Arms composed of multi-tablet regimens were included only if the intervention arm in the study was an STR
- For studies with multiple regimens in the comparison arm, the regimen with the most participants was used
- Outcomes of interest were rates of TE-RAMs and discontinuation due to AEs
- · Risk of bias (RoB) in the RCTs was evaluated using the Cochrane RoB 2 tool to assess internal validity
- Risk ratios (RRs) with 95% CIs were estimated using a random-effects model
- · Surface under the cumulative ranking curve (SUCRA) was used to rank interventions to prevent TE-RAMs and discontinuation due to AEs
- SUCRA scores signal the probability a treatment has of being among the best options in the network; higher scores represent better ranking

Results

· Overall, 19 RCTs encompassing 10,760 participants were included in the analysis (Figure 1)⁶⁻²⁴



Figure 1. PRISMA Flow Chart



Items for Systematic reviews and Meta-Analyses; RCT, randomised controlled trial; STR, single-tablet regimer

- The network map in Figure 2A shows which treatment regimens were compared in the 19 RCTs included in the analysis of rates of TE-RAMs
- Tests for inconsistency (P = 0.999) and residual heterogeneity (I²: 0.00; P = 0.996) were not significant
- The network map in Figure 2B shows which treatment regimens were compared in the 10 RCTs included in the analysis of rates of discontinuation due to AEs
 - Tests for inconsistency (P = 0.621) and residual heterogeneity (I^2 : 0.00; P = 0.368) were not significant

Figure 2. Network Maps of (A) TE-RAMs and (B) Discontinuation Due to AEs at 48 Weeks^{a,b}

- · At 48 weeks, the risk of TE-RAMs with bictegravir/emtricitabine/tenofovir alafenamide (B/F/TAF) and dolutegravir/abacavir/lamivudine (DTG/ABC/3TC) was an estimated 78% lower than that of CAB + RPV every 8 weeks (Q8W; RR, 0.22 [95% CI, 0.02-2.02] and 0.22 [95% CI, 0.00-19.72], respectively) and tended to be lower than that of CAB + RPV every 4 weeks (Q4W) and all 2- and 3-drug STRs (Table 1)
- The risk of TE-RAMs with CAB + RPV Q4W was an estimated 60% lower than that of CAB + RPV Q8W (RR, 0.40 [95% CI, 0.15-1.09])
- There were no statistically significant differences between treatment regimens for TE-RAM RRs
- The risk of discontinuing therapy due to AEs at 48 weeks was significantly (85% and 84%) lower with B/F/TAF compared to CAB + RPV Q4W and CAB + RPV Q8W, respectively (Table 1)
- The risk of discontinuing therapy due to AEs was significantly (95%) lower with DTG/ABC/3TC compared to both CAB + RPV Q4W and CAB + RPV Q8W
- B/F/TAF ranked the highest for probability of preventing TE-RAMs (71.4%), and efavirenz/emtricitabine/tenofovir disoproxil fumarate ranked the lowest (23.1%; Table 2)
- CAB + RPV Q8W showed a lower probability of preventing TE-RAMs than all integrase strand transfer inhibitorand protease inhibitor-based STRs (33.7%)
- Both CAB + RPV regimens ranked the lowest for probability of preventing discontinuation due to AEs (Table 2)

Table 1. Pooled Estimates for Risk of TE-RAMs and Discontinuation Due to AEs at 48 Weeks^a

Table 2. SUCRA Rankings of TE-RAMs and **Discontinuation Due to AEs at 48 Weeks**

		Pooled Estimates (RR [95% Cl])	
Compariso	n Regimen	TE-RAMs	Discontinuation Due to AEs
B/F/TAF /s	EFV/FTC/TDF	0.12 (0.01-2.22)	-
	RPV/FTC/TDF	0.12 (0.00-7.15)	-
	bPI + 2 NRTIs	0.21 (0.02-1.90)	1.46 (0.18-11.49)
	CAB + RPV Q8W	0.22 (0.02-2.02)	0.16 (0.04-0.67)
	E/C/F/TXF	0.34 (0.04-2.61)	2.16 (0.24-19.42)
	D/C/F/TAF	0.41 (0.00-37.45)	1.34 (0.13-13.56)
	DTG/3TC	0.44 (0.01-15.60)	-
	CAB + RPV Q4W	0.54 (0.06-5.27)	0.15 (0.03-0.75)
	DTG/RPV	0.68 (0.01-60.43)	0.69 (0.02-24.91)
	RPV/FTC/TAF	0.84 (0.02-40.27)	-
	DTG + 2 NRTIs	0.99 (0.02-49.69)	0.99 (0.32-3.03)
	DTG/ABC/3TC	1.00 (0.02-50.04)	2.99 (0.61-14.68)
CAB + RPV Q4W vs	EFV/FTC/TDF	0.22 (0.01-4.57)	-
	RPV/FTC/TDF	0.22 (0.00-14.23)	-
	bPI + 2 NRTIs	0.38 (0.03-4.58)	9.85 (0.71-136.53)
	CAB + RPV Q8W	0.40 (0.15-1.09)	1.08 (0.50-2.35)
	E/C/F/TXF	0.62 (0.08-4.72)	14.61 (0.95-224.66)
	D/C/F/TAF	0.76 (0.01-79.17)	9.03 (0.53-153.28)
	DTG/3TC	0.82 (0.02-29.88)	-
	DTG/RPV	1.25 (0.01-127.83)	4.66 (0.09-239.34)
	RPV/FTC/TAF	1.54 (0.03-80.53)	-
	DTG + 2 NRTIs	1.82 (0.02-168.37)	6.69 (0.93-48.22)
	DTG/ABC/3TC	1.83 (0.02-169.56)	20.20 (2.07-196.86)
	B/F/TAF	1.84 (0.19-17.83)	6.76 (1.33-34.42)
CAB + RPV Q8W vs	EFV/FTC/TDF	0.55 (0.02-12.15)	-
	RPV/FTC/TDF	0.55 (0.01-37.28)	-
	bPI + 2 NRTIs	0.95 (0.07-12.16)	9.11 (0.74-112.34)
	E/C/F/TXF	1.55 (0.18-13.16)	13.51 (0.98-185.72)
	D/C/F/TAF	1.91 (0.02-205.02)	8.35 (0.55-127.23)
	DTG/3TC	2.05 (0.05-79.26)	-
	CAB + RPV Q4W	2.51 (0.92-6.90)	0.92 (0.43-2.01)
	DTG/RPV	3.14 (0.03-331.05)	4.31 (0.09-204.95)
	RPV/FTC/TAF	3.87 (0.07-211.48)	
	DTG + 2 NRTIs	4.57 (0.05-415.33)	6.18 (1.00-38.06)
	DTG/ABC/3TC	4.61 (0.05-418.26)	18.68 (2.20-158.91)
	B/F/TAF	4.62 (0.50-43.15)	6.25 (1.49-26.15)

	SUCRA Score (%)		
Treatment Regimen	TE-RAMs	Discontinuation Due to AEs	
B/F/TAF	71.4	51.8	
RPV/FTC/TAF	65.2	-	
DTG + 2 NRTIs	63.9	51.9	
DTG/ABC/3TC	63.7	81.3	
CAB + RPV Q4W	59.1	10.4	
DTG/RPV	58.6	44.7	
DTG/3TC	52.8	-	
D/C/F/TAF	50.3	59.4	
E/C/F/TXF	46.6	76.3	
CAB + RPV Q8W	33.7	12.0	
bPI + 2 NRTIs	33.6	62.3	
RPV/FTC/TDF	27.9	-	
EEV/ETC/TDE	23.1		



ference(s) for each comparison is denoted in red on the network regs. size is proportional to the number of protingents across all included studies for a treatment regimen, and line thickness is proportional to the number of studies that compared the 2 treatment regimes are proportional to the number of protingents across all included studies for a treatment regimes. A series event, BFT/TAF, bictogravit/emtribitabinetendovir alleferantics, bPH, boosted protease inhibitor, CAB + RPV, cabolegavir (https://winter.bDCFT/TAF, duclegavir/emtribitabinetendovir outgeavir, DTGT2C, duclegavir/emtribitabinetendovir discorded in the state across and the state a

rRPs are for this count is determined by the second sec

vert, BF/TAF, biclegravit/entricitabine/tenofou/ralafenamide, PP, 1 biodegravit + ritpivinne, DiCPT/RF, darunavit/cobcistate/entricitabin vertice/strategravit/entrivatione, ICT04/BC3705, colutegravit formandoj, EFV/FTCTDF, elsivienza/entricitabine/tenofov/ diogopor Japasi Inhibitor, QMV, every 4 veeks; QMV, every 5 veeks; RPV/FT amide, RPV/FTCTDF, ritpivine/entricitabine/tenofov/ diogopor amide, RPV/FTCTDF, ritpivine/entricitabine/tenofov/ diogoport amide, RPV/FTCTDF, ritpivine/entricitabine/tenofov/ diosoport amide, RPV/FTCTDF, ritpivine/entricitabine/tenofov/ association association disoprox ite: NRT

Limitations

- · In studies with multiple regimens in the comparison arm, only 1 regimen could be included in the network, limiting the overall dataset
- A low number of events was observed for both TE-RAMs and discontinuation due to AEs
- · There were few direct head-to-head clinical trials in the analysis
- · The majority of studies were open-label studies, which are susceptible to AE reporting bias
- The analysis of discontinuation due to AEs was restricted to 10 out of the 19 total studies due to the lack of details when comparator arms contained multiple regimens

References: 1. US Department of Health and Human Services. Guidelines for the use of antiretroviral agents in adults and adolescents with HIV. Accessed 10 September 2024. https://clinicalinfo.hiv.gov/ stes/default/files/guidelines/documents/adult-adolescent-arv/guidelines-adult-adolescent-arv.pdf. 2. Altice F, et al. Patient Prefer Adherence. 2019;13:475-90. 3. Carr A, et al. Antivir Ther. 2023;28: 13596553231201162. 4. Overton ET, et al. Clin Infect Dis. 2023;76:1646-54. 5. World Health Organization. HIV drug resistance report 2017. Accessed 17 September 2024. https://www.who.int/publications/l/ leum?9829415231. 6. Daar ES, et al. Lancet HIV. 2015;%437-56. 5. S. Averton ET, et al. Clin Infect Dis. 2017;38:455-93. Artise J, et al. H. 2017 Mir Trais. 2017;18:118-25. 10. DeJesus E, et al. Lancet HIV. 2017;4:e195-2015. "UBIS: %437-56. S. Bax HIV. 2015;%437-56. S. Bax HIV. 2017;4:e195-60. S. Bax HIV. 2015;437-56. S. Bax HIV. 2015;43

nts: This study was funded by Gilead Sciences, Inc. Medical writing support was provided by Katherine Townsend, PhD, of Lumanity Communications Inc., and was funded by Gilead Sciences, Inc.

Disclosures: IR. CW. HWS, and NC are employees of the University of Utah, which received a research grant from Gilead Sciences. Inc., to conduct this work: CW also received funding for travel and confere Unscriptions, inc., to control to the University of Data (which incomes a research grant non-block) cale to the Universe in Counter in the Universe in Counter in the Universe in Counter incomes interview of the Universe in Counter incomes income

Correspondence: Nathorn Chaivakunapruk, nathorn.chaivakunapruk@utah.edu